Author's response to reviews

Title: Low concentrations of serum testosterone predict acute myocardial infarction in men with type 2 diabetes mellitus

Authors:

Bledar Daka (bledar.daka@allmed.gu.se)
Robert D Langer (rlanger@medicine.nevada.edu)
Charlotte A Larsson (charlotte_a.larsson@med.lu.se)
Thord Rosen (thord.rosen@medic.gu.se)
Per Anders Jansson (per-anders.jansson@medic.gu.se)
Lennart Råstam (lennart.rastam@me.com)
Ulf Lindblad (ulf.lindblad@gu.se)

Version: 4 Date: 8 July 2015

Author's response to reviews: see over
Dear Editor,

Thank you for the review of our manuscript “Low concentrations of serum testosterone predict acute myocardial infarction in men with type 2 diabetes mellitus”, and the opportunity to resubmit a revised version. We indeed appreciate the constructive criticism raised by the reviewers and we tried to address them as careful as possible.

Below, You will find the answer to the reviewer.

We hope you will find that our revised manuscript meets all issues from you and the reviewers.

Gothenburg, 2015-07-08

Bledar Daka
Reviewer Giulia Rastrelli

Thank you for the comments.

The Authors report that analyses containing estradiol were deleted, however table 3 still reports estradiol. In table 1 and 2 several units of measurement are still lacking.

Answer: We have made the suggested changes on deleting estradiol reporting in table 3 and on adding the units of measurements in table 1 and 2.